Author's response to reviews

Title: Effectiveness of intramuscular corticosteroid injection versus placebo injection in patients with hip osteoarthritis: design of a randomized double-blinded controlled trial

Authors:

Desirée M.J. Dorleijn (d.dorleijn@erasusmc.nl)
Pim A.J. Luijsterburg (p.luijsterburg@erasusmc.nl)
Max Reijman (m.reijman@erasusmc.nl)
Margreet Kloppenburg (G.Kloppenburg@lumc.nl)
Jan A.N. Verhaar (j.verhaar@erasusmc.nl)
Patrick J.E. Bindels (p.bindels@erasusmc.nl)
P. Koen Bos (p.k.bos@erasusmc.nl)
Sita M.A. Bierma-Zeinstra (s.bierma-zeinstra@erasusmc.nl)

Version: 3 Date: 2 December 2011

Author's response to reviews: see over
Dear editorial board of BioMed Central,

Thank you for accepting our manuscript. “Effectiveness of intramuscular corticosteroid injection versus placebo injection in patients with hip osteoarthritis: design of a randomized double-blind controlled trial”, by Desirée M.J. Dorleijn, et al.

You will find the revised manuscript enclosed. The following formatting changes have been made:

1. Abstract: Please amend the abstract so that it fits the following structure: Background, Methods/Design, Discussion.
   Answer: discussion section is amended.

2. Tables: Table 1 is not divided into columns. Please re-format this information so that it is both displayed in rows and columns, with the data held in cells.
   Answer: Table 1 is changed into columns and rows.

3. References: If reference 16 has been editorially accepted, please list it as ‘in press’.
   Please note that references cannot be in the reference list if they are no editorially accepted/published.
   Answer: this reference has recently been published, publication date inserted.

4. Emails: Please provide email address for authors: Luijsterburg, Reijman, Kloppenburg, Verhaar, Bindels, Bos, Bierma-Zeinstra.
   Answer: email addresses have been provided for the above mentioned authors on the title page.

5. Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.
   Answer:
   1. Abstract
      Methods/design section: orthopedic surgeon changed to orthopaedic surgeon
   2. Methods, section patient selection
      Orthopedic changed to orthopaedic
   3. Methods, section randomisation
Orthopedics changed to orthopaedics
4. Methods, section Questionnaires at baseline, 2, 4, 6 and 12 weeks
   Diary changed to diary

6. The affiliation of Reijman, Kloppenburg, Verhaar and Bos is not Aff1.
   We have changed this into the correct affiliations on the title page:
   affiliation Max Reijman, Aff2
   affiliation Margreet Kloppenburg, Aff3
   affiliation Jan AN Verhaar, Aff2
   P Koen Bos, Aff2
   Aff 2 = Department of Orthopaedics, Erasmus MC, University Medical Center,
   Rotterdam, PO Box 2040, 3000 CA Rotterdam, The Netherlands
   Aff 3 = Department of Rheumatology, LUMC, PO Box 9600, 2300 RC Leiden, the
   Netherlands

Sincerely yours,

Desirée M.J. Dorleijn, on behalf of the authors.

Corresponding author: Desirée M.J. Dorleijn at Erasmus MC, University Medical Center, PO
Box 2040, 3000 CA, Rotterdam, the Netherlands, d.dorleijn@erasmusmc.nl
phone number: +31107044196, fax number: +31107032127.